Author’s response to reviews

Title: LMA® GASTRO™ Airway for endoscopic retrograde cholangiopancreatography: a retrospective observational analysis

Authors:

Andre Tran (andre.tran24@gmail.com)
Venkatesan Thiruvenkatarajan (Venkatesan.Thiruvenkatarajan@sa.gov.au)
Medhat Wahba (Medhat.wahba@sa.gov.au)
John Currie (John.Currie@sa.gov.au)
Anand Rajbhoj (arajbhoj@gmail.com)
Roelof van Wijk (Roelof.VanWijk@sa.gov.au)
Edward Teo (eteoaddress@yahoo.com.au)
Mark Lorenzetti (marklorenzetti@internode.on.net)
Guy Ludbrook (guy.ludbrook@sa.gov.au)

Version: 3 Date: 17 Apr 2020

Author’s response to reviews:

Editor comment 1 re: administrative consent
- The data was access as per ethics approval. This study was reviewed and approved by the Central Adelaide Local Health Network Human Research Committee. The Chairman of the ethics committee granted permission to access the medical records. Our ethics application to be approved required detailed descriptions of how the data would be extracted, stored and analysed. Informed consent of participants in this retrospective observational analysis was not sought and justified.

Editor comment 2 re: Figure 1
- The figure legend has been updated. It writes the image in Figure 1 was obtained and adapted from Teleflex, Australia with written permission.